Author’s response to reviews

Title: Occipital lobe infarction: a rare presentation of bilateral giant cavernous carotid aneurysms: a case report

Authors:

Kavin Vanikieti (Vanikieti.kavin@gmail.com)
Anuchit Poonyathalang (Au.tumn@yahoo.com)
Panitha Jindahra (panitha.jin@mahidol.ac.th)
Piyapho Cheecharoen (Cheecharoen.neurorama@gmail.com)
Wimonwan Chokthaweesak (Bellz_zone@yahoo.com)

Version: 1 Date: 05 Jan 2018

Author’s response to reviews:

To editor,

We have edited the language and grammar with the help of native speakers and some as you suggested.

Please kindly see the revised manuscript.

Thank you very much.